Math 132 Winter 2004 R. Schultz

UPDATED GENERAL INFORMATION — FEBRUARY 23, 2004

Here is the fifth homework assignment, which is due in class on Monday, March 2, 2004. “F&B” refers to the course text by Fraleigh and Beauregard.

Exercises 8, 12, 18 and 20 beginning on p. 499 of F&B (for the last two, only give a Jordan form for the matrix).

Additional Exercise 1 for Section IV.4

Additional Exercise 1 for Section IV.5

Review information regarding the second midterm examination will be included in the next update.